Author's response to reviews

Title: Pre-hospital ECG for acute coronary syndrome in urban India: A cost-effectiveness analysis

Authors:

Joshua Schulman-Marcus (Joshua.Schulman-Marcus@mssm.edu)
Dorairaj Prabhakaran (dprabhakaran@ccdcindia.org)
Thomas Gaziano (tgaziano@partners.org)

Version: 4 Date: 1 March 2010

Author's response to reviews:

Dear Editors,

We have made the requested changes in the revised manuscript.
Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.

We changed "Conclusions" to "Discussion and Conclusions".
Table citations - we made the correct changes in the table citations to match the order in the manuscript.

Minor revisions

We removed the footer.

Title page - we removed word and figure count and keywords. Funding information should be in the Acknowledgements section.

Authors' contributions - we moved to a position after the Conclusions.

Underlining was removed from the text.

Tables- removed the visible vertical lines from your tables.

Figures: We created a Table and Figure legend and are uploading the two figures separately.

Sincerely,

Thomas A Gaziano